Author's response to reviews

Title: Occult pneumothorax, revisited

Authors:

Hesham R. Omar (hesham_omar2003@yahoo.com)
Devanand Mangar (dmangar@tgh.org)
Rania Rashad (rania185@hotmail.com)
Engy Helal (helal_engy@yahoo.com)
Enrico M. Camporesi (ecampore@health.usf.edu)

Version: 9 Date: 2 September 2010

Author's response to reviews:

Dear Editor,

minor language corrections has been made to the manuscript for better flow of the text

Sincerely,

Hesham R. Omar